Author's response to reviews

Title: Impact of HIV vaccination on laboratory diagnosis: case reports.

Authors:

Suthon Vongsheree (suthon@dmsc.moph.go.th)
Archawin Rojanawiwat (aechawim191@yahoo.com)
Chanchai Chardbanchachai Chardbanchachai (chadbuncha@hotmail.com)
John Lerwitworapong (john@samarts.com)
Wichuda Kongpromsook (kvichuda@yahoo.com)
Wiroj Paungtubtim (wirojptt@yahoo.com)
Hansa Thaisri (hansat@dmsc.moph.go.th)
Pathom Sawanpanyalert (pathoms@loxinfo.co.th)
Pongnuwat Sri-ngam (p_nuwat@hotmail.com)
Silaporn Pithak (kungfumail@yahoo.com)
Wimala Inunchot (wimalain@yahoo.com)

Version: 5 Date: 10 Jul 2002

Dear Editors
I corrected some details of the manuscript according to your suggestion.
The content of the manuscript is the same as the original one that I have submitted.
Regard,
Suthon Vongsheree